Author’s response to reviews

Title: Malaria Rapid Diagnostic Tests in fever case management: a study of the experience with their use at lower level health centres in Uganda

Authors:

Caroline Asiimwe (caroline.asiimwe@finddiagnostics.org)
Daniel J Kyabayinze (d.kyabayinze@malariaconsortium.org)
Zephania Kyalisiima (zkyaliisiima@ss.mak.ac.ug)
Jane Nabakooza (jinksuganda@yahoo.com)
Moses Bajabaite (cmbaja@yahoo.com)
Helen Counihan (h.counihan@malariaconsortium.org)
James K Tibenderana (j.tibenderana@malariaconsortium.org)

Version: 10 Date: 2 January 2012

Author’s response to reviews:

Dear Editorial Team,

this is to confirm that we received and reviewed comments and suggested changes from your team. We have addressed comments #1 and 2 regarding physical addresses of the authors. The suggested changes/edits in the rest of the document look very reasonable. We have accepted all edits.

Asiimwe Caroline (First author)